Author's response to reviews

Title: Information Transfer: What do decision-makers want and need from researchers

Authors:

- Maureen Dobbins (dobbinsm@mcmaster.ca)
- Peter Rosenbaum (rosenbau@mcmaster.ca)
- Nancy Plews (plewsn@mcmaster.ca)
- Mary Law (lawm@mcmaster.ca)
- Adam Fysh (fysha@mcmaster.ca)

Version: 5 Date: 29 June 2007

Author's response to reviews:

Dear Production team, I have addressed the following editing issues:

- included county of affiliation for all authors
- ensured submission abstract and manuscript abstract are identical
- provided assurance that all authors have read the final manuscript
- removed the one duplicate citation